Author's response to reviews

Title: Modern contraceptive use among sexually active men in Uganda: Does discussion with a health worker matter?

Authors:

Allen Kabagenyi Ms. (allenka79@yahoo.com)
Patricia Ndugga Ms. (patnduggab@yahoo.com)
Stephen Ojiambo Wandera Mr. (swandera@gmail.com)
Betty Kwagala Dr. (kkwagala@gmail.com)

Version: 3 Date: 29 July 2013

Author's response to reviews: see over
Dear Editor,

**RE: Revision for Manuscript 1251652065989072**

Thank you for accepting our manuscript and for permitting us to make some changes as suggested. The paper entitled “Modern contraceptive use among sexually active men in Uganda: Does discussion with a health worker matter?”

1. We have revised the manuscript and made modifications in the methodology section, and we indicate that we used secondary data obtained from Measure DHS website after seeking for approval for acquisition (more details on sample selection are described in [9]). *Uganda Demographic and Health Survey* (UDHS) 2011 data was freely accessed from the MEASURE DHS website. Authorization to use this data was obtained upon providing a brief description of our study. Demographic and Health Survey (DHS) dataset are in the public domain and are permitted to be used on request.

2. Informed consent; consent for participation was sought as per the introduction and consent forms attached in the appendix section of the Demographic and Health Survey report, before the household schedule.

3. Authors’ Information section was removed as suggested.

Thank you, looking forward to your kind consideration

Allen k  
on behalf of the co-authors